

Delft University of Technology

Neuroanatomical anomalies due to a defect in the FGF3 gene, associated with the Labyrinthine Aplasia, Microtia and Microdontia syndrome

insights from the placement of auditory brainstem implants in two siblings

Frijns, Johan H.M.; Geerders, M. G.S.; Scholing, Esther; Verbist, Berit M.; Koot, Radboud W.; Malessy, Martijn J.A.; Boermans, Peter Paul B.M.; Briaire, Jeroen J. **DOI**

10.1177/26330040241290834

Publication date 2024 Document Version Final published version Published in Therapeutic Advances in Rare Disease

Citation (APA)

Frijns, J. H. M., Geerders, M. G. S., Scholing, E., Verbist, B. M., Koot, R. W., Malessy, M. J. A., Boermans, P. P. B. M., & Briaire, J. J. (2024). Neuroanatomical anomalies due to a defect in the FGF3 gene, associated with the Labyrinthine Aplasia, Microtia and Microdontia syndrome: insights from the placement of auditory brainstem implants in two siblings. *Therapeutic Advances in Rare Disease*, *5*. https://doi.org/10.1177/26330040241290834

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Ther Adv Rare Dis

2024, Vol. 5: 1–9

permissions

26330040241290834 © The Author(s), 2024. Article reuse guidelines: sagepub.com/journals-

Neuroanatomical anomalies due to a defect in the FGF3 gene, associated with the Labyrinthine Aplasia, Microtia and Microdontia syndrome: insights from the placement of auditory brainstem implants in two siblings

Johan H. M. Frijns, (Roos) M. G. S. Geerders, Esther Scholing, Berit M. Verbist, Radboud W. Koot, Martijn J. A. Malessy, Peter-Paul B. M. Boermans and Jeroen J. Briaire

Abstract: Here, we describe two congenitally deaf male siblings with the same compound heterozygotic, likely pathogenic mutations in the FGF3 gene, associated with the labyrinthine aplasia, microtia and microdontia (LAMM) syndrome. Both children had bilateral cochleovestibular aplasia, precluding cochlear implantation. The elder brother received an auditory brainstem implant (ABI) with very limited auditory responses. During the ABI-surgery of the younger subject, it was discovered that excellent auditory responses could be obtained when the electrode array was placed considerably more caudally and more medially than standard. It was observed that the foramen of Luschka, the entrance to the lateral recess of the fourth ventricle was located more caudally. In view of this observation the good auditory development of the latter child, it was decided to give the older child a contralateral ABI. Again, it turned out that the anatomy of the brainstem was abnormal with a more caudal location of the foramen of Luschka and the cochlear nucleus, and this child is showing good progress with his auditory development. It is concluded that one should be aware of the anatomical differences at the level of the brainstem when placing an ABI in children with this genetic disorder (and most likely also in the LAMM syndrome). This also underpins the need of a multidisciplinary approach with closely collaborating team members and good family guidance when diagnosing and treating children with rare deafness.

Plain language summary

Deviant surgical anatomy for auditory brainstem implants with FGF3 gene defects, associated with LAMM syndrome

This case series describes three placements of Auditory Brainstem Implants (ABIs) in two congenitally deaf brothers with a gene defect associated with the Labyrinthine Aplasia, Microtia and Microdontia (LAMM) syndrome. In line with this syndrome, they both had no inner ears and no hearing nerves and therefore couldn't be helped with a cochlear implant. The only option was an ABI, with a mesh electrode positioned on the cochlear nucleus, the first relay station of the hearing system in the brainstem. The first ABI electrode in the elder brother was placed in the classical position and did not yield much benefit. During

Correspondence to: Johan H. M. Frijns Department of

Otorhinolaryngology & HNS, Leiden University Medical Centre, P.O. Box 9600, Leiden, 2300 RC, The Netherlands

Bioelectronics Group, Faculty of EEMCS, Delft University of Technology, Delft, The Netherlands j.h.m.frijns@lumc.nl

(Roos) M. G. S. Geerders Esther Scholing Peter-Paul B. M. Boermans Jeroen J. Briaire Department of Otorhinolaryngology & HNS, Leiden University Medical Centre, Leiden, The Netherlands

Berit M. Verbist

Department of Radiology, Leiden University Medical Centre, Leiden, The Netherlands

Radboud W. Koot Martijn J. A. Malessy

Department of Neurosurgery, Leiden University Medical Centre, Leiden, The Netherlands

journals.sagepub.com/home/trd



Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (http://www.creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the Sage and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

the ABI surgery in the younger sibling, it was discovered that the location of the cochlear nucleus (the target site for an ABI) was located further down and the electrode had to be inserted deeper. This child showed a good auditory development, and it was decided to give the older child an ABI at the other side. Again, it turned out that the anatomy of the brainstem was abnormal with a lower location of the cochlear nucleus. With this second implant, this child's auditory development is showing good progress. It is concluded that one should be aware of the anatomical differences at the level of the brainstem when placing an ABI in children with this genetic disorder (and most likely also in the LAMM syndrome).

Keywords: ABI, auditory brainstem implant, auditory nerve dysplasia, cochleovestibular dysplasia, FGF3 gene, LAMM-syndrome

Received: 19 March 2024; revised manuscript accepted: 24 September 2024.

Introduction

The FGF3 gene,¹ located on chromosome 11, encodes for fibroblast growth factor 3. During antenatal development, this protein triggers the processes leading to the development of the inner ears (cochleae). Mutations in both alleles of the FGF3 gene are therefore known to cause profound deafness due to severe forms of cochleovestibular dysplasia, including labyrinthine aplasia or Michel's deformity.² Apart from this, FGF3 deficits can lead to concomitant malformations of the outer ear and small and widely spaced teeth, and then it is called the labyrinthine aplasia, microtia and microdontia (LAMM) syndrome.³⁻⁸ This is an extremely rare autosomal recessive disorder, for which no prevalence is known to date, and for which a recent literature review reported 24 pathogenic and likely pathogenic variants of the FGF3 gene.3

The complete absence of inner ear structures in labyrinthine aplasia precludes rehabilitation with cochlear implants (CIs), devices that directly stimulate the auditory nerve with an electrode array, placed in the scala tympani of the cochlea. Therefore, the only available option to provide sound perception, and possibly the development of oral communication, is by using an auditory brainstem implant (ABI).⁹ This device technically resembles a CI, but the electrode contacts are on a mesh, placed on the cochlear nucleus, the first relay station of the auditory pathway in the brainstem.

This article describes the rehabilitation with ABIs of two brothers (indicated as Child A and B) with compound heterozygotic mutations in the FGF3

gene in our tertiary referral centre. Apart from bilateral labyrinthine aplasia, they both had bilateral aplasia of the auditory nerve, making these children very suitable candidates for rehabilitation with ABIs. To our knowledge, there is very limited literature on the use of ABIs in cases with FGF3 mutations, including the LAMM syndrome. We are only aware of one case, included in a recent publication on ABI results in 38 children and a review of the literature.¹⁰The focus will be on the consequences for the electrode placement of the anatomical abnormalities found in the brainstem in both siblings, which may need special consideration when performing ABIsurgery in this patient population.

Auditory brainstem implants

An ABI is a specialized medical device designed to restore a sense of sound to individuals who are profoundly deaf due to non-functional or absent cochlear nerves, conditions that preclude the use of conventional hearing aids or CIs. Unlike CIs, which stimulate the cochlea in the inner ear directly to convey sound signals to the brain via the cochlear nerve, ABIs bypass the cochlea and the cochlear nerve entirely. They work by directly stimulating the cochlear nucleus (CN) in the brainstem, the part of the auditory pathway responsible for processing auditory information received from the cochlea. The CN is located in the floor of the lateral recess of the fourth ventricle (Figure 1), which is surgically accessible via a neurosurgical route, the so-called 'retro-sigmoidal' approach. In our centre, the procedure is performed by an



Figure 1. (a) Top view of the brainstem at the level of the fourth ventricle. The dashed arrow indicates the route of insertion of an ABI via the foramen of Luschka into its lateral recess, guided by the course of the glossopharyngeal nerve (IX). In blue, the cochlea, auditory nerve (VIII) and the CN are indicated. In the cases described in this article, both the cochleae and auditory nerves were absent, creating the indication for an ABI. (b) Detail of the foramen of Luschka, the route of access to the CN, which is usually located in the extension of the IXth cranial nerve, just medial (left in the figure) to Sollman's vein.

Source. Adapted (with permission) from Rosahl.¹¹

auditory brainstem implant (ABI); CN, cochlear nucleus.

ENT-surgeon and a neurosurgeon who both have ample experience with this procedure, guided by electrophysiological measurements (electrically evoked auditory brainstem responses, eABRs) by two clinical physicists who have both ample experience with such recordings. All implantations described in this study were performed with a MED-EL Mi1200 SYNCHRONY PIN ABI implant. The intraoperative measurements were performed with a Natus Nicolet EDX (Natus Medical, Orlando, FL, USA) recording device. For the measurements, aimed to find the right position on the brainstem, the so-called 'ABI Placing Electrode' from MED-EL was used. The placing electrode has two advantages compared to the electrode array of the ABI implant. First, it has a smooth surface, without the Dacron mesh used for stabilization of the final electrode array, reducing the chance of damaging the brainstem when it is moved to find the right position. Second, the electronics are located outside the patient, reducing stimulus artefacts.

Sennaroglu et al.¹² were among the first to demonstrate the potential role of ABIs in providing auditory rehabilitation to prelingually deaf children with inner ear malformations, including severe stenosis of the cochlear aperture and cochlear nerve aplasia, as occurring in individuals with FGF3 gene defects (and the LAMM syndrome). Additionally, Monsanto et al.13 offered insights into the outcomes of ABIs in children based on a review of the literature, providing further understanding of the efficacy and challenges associated with ABI use in populations.¹³ The oral-aural speech and language development with ABIs in congenitally deaf children requires a much more intensive and longer lasting rehabilitation process than with CIs. The procedures used in our centre, and the differences between both programmes are the same as the ones described in detail by Van der Straaten et al.,14 who also made a comparison between the results with CI and ABI in our centre, especially in congenitally deaf children with additional handicaps.

Paediatric rehabilitation programme for ABIs

Children are considered for an ABI when imaging reveals that the auditory nerve and/or cochlea are poorly developed, or in cases of congenital malformation or aplasia of the cochlea, cochlear trauma, or cochlear ossification following meningitis. The Leiden University Medical Centre is the sole hospital in the Netherlands performing ABI surgeries for children. The ABI rehabilitation programme at our clinic is an intensified extension of our CI rehabilitation programme and consists of so-called 'intensive weeks' with fixed components. However, there is a significant difference in duration and intensity of the programme for CI and ABI recipients.

A general basis for the rehabilitation for children with an ABI consists out of 12 rehabilitation weeks spread over the first 2 years post-implantation, compared to 6 weeks for CI children, distributed over the first year after implantation. Research and clinical experience suggest that children with an ABI require more and longer rehabilitation due to the complexity of their condition. An intensive week includes three to four mornings of appointments with a speech therapist, audiologist and psychologist. The goal of this specialized treatment programme is to adjust the ABI and integrate its use into the child's world and his surroundings as effectively as possible. Each intensive week follows a structured approach, including fittings (ABI tuning), auditory stimulation (focusing on expanding various phases of auditory development, encouraging vocalization through play and sometimes spoken language development), Pedagogical Interaction Guidance (using play materials to work on communicative prerequisites like eve contact, turntaking, concentration or other areas of focus) and auditory testing with the ABI. The exact structure and intensity is tuned to the specific child, its development and the support system. We advise the parents to use sign-supported language as the preferred mode of communication with their child, as it will usually take many years for the ABI to yield sufficient oral-aural communication for use in daily life. In addition, all parents and children get education in sign language via the home support service (or schools for the deaf).

Early language development is monitored with validated questionnaires, including the IT-MAIS-NL,¹⁵ a validated Dutch translation of the

Infant-Toddler Meaningful Auditory Integration Scale,¹⁶ resulting in a score between zero and 40, and the Categories of Auditory Performance (CAP),¹⁷ a scale on which children's developing auditory abilities can be rated in the following eight categories in order of increasing difficulty, ranging from (CAP=0) 'displays no awareness of environmental sounds' to (CAP=7) 'can use the telephone with a familiar talker'.

Presentation of cases

Case A: Management and rehabilitation

Patient A was a neonate, referred to a tertiary audiological/CI centre for further evaluation following a negative outcome from the neonatal hearing screening, with the initial consideration for rehabilitation with a CI. In addition, he had bilateral ear pits, a right-sided cup ear, a mild dolichocephaly (elongated head shape), a preaxial oligodactyly of the left thumb and was slightly hypotonic. He had no additional needs (no cognitive impairment, normal vision). Subsequent radiological assessments, including computed tomography (CT), revealed the absence of both cochleae, indicating complete labyrinthine aplasia and the bilateral absence of the cochlear nerve (Figure 2). Concurrent with these findings, genetic testing was undertaken to provide additional insights into the patient's condition. It turned out that he had the following compound heterozygotic, likely pathogenic mutations in the FGF3 gene:

 NM 005247.2(FGF3):c.283C>T, p.(Arg95Trp), maternally
NM 005247.2(FGF3):c.45deIC, p.(Trp16fs), paternally

From the moment, it was confirmed that the child was deaf, his parents began learning Dutch Sign Language (NGT) at home, making NGT and Sign Supported Dutch (NmG) their primary means of communication. The mother is employed as a childcare worker, and the father works as a service technician.

Given the absence of both cochleae and cochlear nerve, the patient was referred to our expert centre for rare ear diseases for an ABI evaluation at 4 months of age. The ABI surgery was performed on the left side when the patient was 14 months



Figure 2. (a1) Axial CT scan of right ear of normal 1-year old child at the level of the cochlea. (a2) Axial CT scan of the same ear as in A1, now at the level of lateral semicircular canal and vestibule. (a3) Reoriented CT scan of the same ear as in A1 and A2, now at the level of the internal auditory canal. (b) Axial CT scan of child A, demonstrating the bilateral absence of the cochlea, semicircular canals and internal auditory canal, which is also not visible in other slices.

old. During the surgical procedure, the posterior fossa was small, especially in the basal direction and the complete absence of the vestibulocochlear nerve (n. VIII) was confirmed. The position of the CN was not easily found, and the placing electrode was tested in four different positions. At high current levels, one electrode pair elicited clear responses with latencies indicative of auditory stimulation. However, in several locations, a later response (4–6 ms), not associated with auditory stimulation, was observed. Based on these bipolar recordings, it was concluded that the CN was small in this case and the final electrode position was determined in such a way that the CN was expected to be in the centre of the array. With the implant set in the monopolar stimulation mode (which is the way the implant is intended to be used and which requires an irreversible action during surgery in this implant), three contacts elicited clear auditory responses, with a possible fourth showing similar outcomes. However, the stimulation levels needed to elicit these responses were much higher than usual. Other contacts elicited either no responses or responses with longer latency that was non-auditory in nature.

Throughout the rehabilitation process, it became necessary to deactivate most of the contacts due to the nature of the responses observed. Ultimately, only three contacts remained active, with stimulation provided at maximum tolerable levels. Free field testing revealed that the patient exhibited very minor reactions to acoustic stimuli, suggesting a limited but discernible capacity for auditory perception through the ABI. Hearing development remained low as shown in Figure 3 where he IT-MAIS-NLs score remained at around 3 points of a 40-point scale and CAPscore remained 0 indicating that there was no awareness of environmental sounds. Nevertheless, the child continued to wear his speech processor on a daily basis and his mother noticed that he produced more variable utterances when the implant was active.

Case B: Implantation and rehabilitation of the younger brother

Child B's hearing journey paralleled that of his older sibling, child A, only with foreknowledge of the condition. He was born when child A was



Figure 3. Hearing development over time scores of both siblings, with the IT-MAIS, which has a maximum score of 40. Data of both children are plotted against the time from hookup of their individual devices, in months. Surgery of child A was at age 14 months, the implantation of child B was at the age of 13 months, the second implantation of child A was at age 61 months.

IT-MAIS, Infant-Toddler Meaningful Auditory Integration Scale.

2 years and 7 months old. Following referral from neonatal screening, directly to our centre, the absence of responses on auditory brainstem response recordings and subsequent genetic testing revealed a shared genetic hearing loss with his brother. He had a right-sided limited microtia with an ear-pit and a pointed pinna of the left ear. By the age of 4 months, imaging studies confirmed child B also had complete labyrinthine aplasia and a bilateral absence of the cochlear nerve. Like his older brother, he had no additional needs.

Despite the modest outcomes observed in Child A, the parents decided to proceed with an ABI for Child B, driven by hope and the desire to explore all possibilities for auditory improvement. At the age of 13 months, Child B underwent surgery for ABI placement on the left side. Again, the complete absence of the VIIIth cranial nerve was confirmed during surgery. In this case, it became clear that the choroid plexus (see Figure 1) was abnormally located, caudally to the IXth nerve, but still marking the entrance to the lateral recess of the fourth ventricle. As a result, the placing electrode was positioned in a non-standard location during the procedure. The initial placement, which was relatively deep into the lateral recess, yielded very positive results, with clear auditory responses detected across all bipolar combinations. The success at this unusual location led to its selection for the final electrode placement,

where seven electrodes demonstrated clear auditory responses at relatively low stimulation levels. The other contacts elicited responses at higher levels.

The progress at rehabilitation for Child B has been promising, with regular adjustments made to increase levels and optimize auditory input. Some electrode produced non-auditory stimulations such as arm movement, head movement and disbalance, leading to their deactivation. The remaining seven active contacts provided auditory percepts, which was further objectified during free field-testing. Child B has shown good responses to musical instruments, though warble tones gained less of his attention. Unfortunately, he refuses to wear his speech processor all day and is not too interested in sound. Auditory thresholds have been identified at around 40 dB HL, marking a significant milestone in Child B's auditory rehabilitation. In Figure 3 the IT-MAIS-NL data of Child B are plotted, showing an increase up to a score of 11 on a 40-point scale after 2 years of ABI use. At that time, his CAP score was 1, that is, awareness of environmental sounds. He clearly outperformed his older brother, and this initiated a new view of case A.

Case A: Implantation of the contralateral side

Considering child B's successful outcomes and the less-than-expected performance of child A,

the medical team revisited child A's case for further evaluation. The intra-operative evoked auditory brainstem responses (eABR) recordings, the fitting outcomes, the initial auditory responses and rehabilitation outcomes for child A all had not met the anticipated objectives. Child B's surgery had led to the hypothesis that the brainstem anatomy in this specific genetic disorder was abnormal, and that it might well be that in child A, the electrode was placed latero-cranially to the CN. Inspired by the positive findings in Child B's case and after a shared decision-making process with the parents, we decided to explore child A's contralateral (right) ear, to see whether an alternative implant location would vield better results for the older sibling.

Child A underwent the procedure to implant the contralateral ear at the age of 5 years and 2 months. The surgical team employed a strategy similar to the one used for Child B. The IXth cranial nerve turned out to be relatively large, and the foramen of Luschka was located extremely caudally, even more than in his sibling. The placing electrode was placed in three different locations, ultimately demonstrating good to acceptable electrically eABRs at standard stimulus levels, with minimal late, non-auditory neural responses. With the final electrode placement, auditory responses were successfully elicited in monopolar mode from all contacts, although two contacts necessitated higher current levels. Notably, five contacts exhibited some non-auditory components in the neural response at elevated levels.

The rehabilitation process (again, following the standard 12 week schedule, spread over 2 years) with the second ABI revealed distinct advantages owing to child A's age. He was already communicative through sign language and thanks to the first ABI possessed a foundational understanding of the concept of hearing, which significantly facilitated the fitting process. Given the very limited benefit of the first implant and the very high current levels, the rehabilitation of the second implant was done while the left-sided implant was not active anymore. Some contacts of the new implant, while not audibly perceptible to him, induced a tingling sensation in his leg and were subsequently deactivated. The rehabilitation process in the first 8 months has shown promising progress, with all active electrodes providing auditory input. Now, he is very motivated to wear his processor all day, and even motivates his

younger brother to do this as well. In Figure 3, the IT-MAIS-NL data of Child A's second implant are plotted, showing IT-MAIS 11 (on a 40-point scale) and a CAP-score 1 meaning awareness of environmental sounds.

Discussion

Here, we present the successful rehabilitation with ABIs of two brothers with compound heterozygotic pathogenic mutations in the FGF3 gene. The review articleby Jamshidi et al.3 on the LAMM syndrome showed that in homozygotic cases of both gene mutations (as found in these siblings) labyrinthine aplasia occurred, which was also the case here. The brothers, however, do not exhibit the full LAMM syndrome as both children had just mild and single-sided microtia. However, both have complete labyrinthine aplasia and the bilateral absence of the cochlear nerve, and both turned out to have microdontia (this was not clear at the time of presentation as neonates as they did not have teeth yet). The oligodactyly in child A (for which he underwent a separate surgery) is not considered a part of the syndrome.

The most interesting observation in this study is the fact that the anatomy of the brainstem is abnormal with this gene defect, and therefore is probably a previously unknown aspect the LAMM syndrome. In fact, we could not find any literature on ABI in children with the LAMM syndrome, and to our knowledge, the fact that the CN is located more caudally and deeper into the lateral recess of the fourth ventricle has not been observed before. This finding was substantiated by the good responses to the atypically placed ABI electrode arrays in child B, the second surgery in child A, and, in retrospect, by the fact that the only three electrode contacts with auditory responses in child A's first implant were located in the caudo-medial portion of the array. We made preoperative MRIs of the whole brain (not shown here), -not only in retrospect-but also could not identify the anatomical differences found during surgery. Although they turned out to be very important determinants of the clinical outcome, the anomalies are, apparently, too subtle to be noticed in imaging.

These cases underscore the importance of individualized care and the potential benefits of reassessing and adjusting treatment strategies based on familial patterns and patient outcomes. For this, it is indispensable to have a multidisciplinary team, which members who collaborate closely in the pre-, peri- and post-operative period. In our centre, we have ample surgical and rehabilitation experience with both CIs and ABIs, as well as with acoustic neuroma/skull base surgery. Thanks to the long-standing collaboration between the members of the surgical team who have jointly performed numerous ABI-surgeries, viz., the ENT- and neurosurgeon, supported by the clinical physicists-audiologists, we were able to identify the pattern of abnormal brainstem anatomy in this syndrome and successfully rehabilitate both children, especially child A after the second surgery.

Conclusion

An abnormally caudal position of the CN in the brainstem was found in two congenitally deaf siblings with cochleovestibular aplasia due to mutations of the FGF3 gene, which is associated with the LAMM syndrome. It is important to be aware of this anatomical abnormality in future cases with this genetic disorder, as correct placement of the electrode is mandatory for a good result with an ABI.

Declarations

Ethics approval and consent to participate

Our institution does not require ethical approval for reporting individual cases or case series. The parents have given written consent for the use of their children's medical data

Consent for publication

The parents have given written consent for the publication of their children's medical data

Author contributions

Johan H. M. Frijns: Conceptualization; Formal analysis; Investigation; Methodology; Supervision; Validation; Visualization; Writing – Original Draft; Writing – Review & Editing.

(Roos) M. G. S. Geerders: Investigation.

Esther Scholing: Data curation; Investigation; Visualization.

Berit M. Verbist: Data curation; Investigation; Visualization.

Radboud W. Koot: Investigation; Visualization; Writing – Original Draft.

Martijn J. A. Malessy: Writing – Review & Editing.

Peter-Paul B. M. Boermans: Data curation; Investigation.

Jeroen J. Briaire: Investigation; Writing – original draft.

Acknowledgements

The authors wish to thank the parents of the children for their permission to describe and publish their medical history, all team members for their dedication and Dr. R.K. Kalkman for his help in preparing the final draft of the manuscript.

Funding

The authors received no financial support for the research, authorship, and/or publication of this article.

Competing interests

The authors declare that there is no conflict of interest.

Availability of data and materials

These are all in the medical records and therefore not available.

ORCID iDs

Johan H. M. Frijns D https://orcid.org/0000-0002-1180-3314

(Roos) M. G. S. Geerders D https://orcid. org/0009-0003-1425-8096

Berit M. Verbist D https://orcid.org/0000-0002-1010-2583

Jeroen J. Briaire D https://orcid.org/0000-0003-4302-817X

References

- MedlinePlus. FGF3 gene, https://medlineplus. gov/genetics/gene/fgf3 (2012, accessed 13 March 2024).
- Nakajima Y. Signaling regulating inner ear development: cell fate determination, patterning, morphogenesis, and defects. *Congenit Anom* (*Kyoto*) 2015; 55: 17–25.

- 3. Jamshidi F, Shokouhian E, Mohseni M, et al. Identification of a homozygous frameshift mutation in the FGF3 gene in a consanguineous Iranian family: first report of labyrinthine aplasia, microtia, and microdontia syndrome in Iran and literature review. *Mol Genet Genomic Med* 2023; 11: e2168. 20230319.
- 4. Al Yassin A, D'Arco F, Morin M, et al. Three new mutations and mild, asymmetrical phenotype in the highly distinctive LAMM syndrome: a report of eight further cases. *Genes (Basel)* 2019; 10: 20190712.
- Basdemirci M, Zamani AG, Sener S, et al. LAMM syndrome: two new patients with a novel mutation in FGF3 gene and additional clinical findings. *Clin Dysmorphol* 2019; 28: 81–85.
- 6. Pathivada L, Krishna MK and Rallan M. A case of lacrimo-auriculo-dento-digital syndrome with multiple congenitally missing teeth. *Case Rep Dent* 2016; 2016: 8563961.
- Ramsebner R, Ludwig M, Parzefall T, et al. A FGF3 mutation associated with differential inner ear malformation, microtia, and microdontia. *Laryngoscope* 2010; 120: 359–364.
- 8. Putri IL, Stephanie A, Pramanasari R, et al. The role of genetic factors in microtia: a systematic review. *F1000research* 2023; 11: 537.
- Sennaroglu L, Sennaroglu G, Yucel E, et al. Long-term results of ABI in children with severe inner ear malformations. *Otol Neurotol* 2016; 37: 865–872.
- Behr R, Schwager K and Hofmann E. Auditory brainstem implants-hearing restoration in congenitally deaf children. *Dtsch Arztebl Int* 2022; 119: 135–141.

- Rosahl SK. Neuroanatomy and physiology relevant to auditory brainstem implants. In: Wilkinson EP and Schwartz MS (eds) *Auditory brainstem implants*. New York: Thieme Medical Publishers; 2021: 3–13.
- 12. Sennaroglu L, Ziyal I, Atas A, et al. Preliminary results of auditory brainstem implantation in prelingually deaf children with inner ear malformations including severe stenosis of the cochlear aperture and aplasia of the cochlear nerve. *Otol Neurotol* 2009; 30: 708–715.
- Monsanto RdC, Bittencourt AG, Neto NJB, et al. Auditory brainstem implants in children: results based on a review of the literature. *J Int Adv Otol* 2015; 284–290.
- van der Straaten TFK, Netten AP, Boermans P, et al. Pediatric auditory brainstem implant users compared with cochlear implant users with additional disabilities. *Otol Neurotol* 2019; 40: 936–945.
- 15. Schaaij-Gulpen PMB, Beers M, De Raeve L, et al. Nederlandstalige meetinstrumenten voor het evalueren van de auditieve en de spraakontwikkeling van kinderen met een Cochleair Implantaat; samenstelling en voorbeelden uit de praktijk. (Dutch tools to evaluate the auditory and speech development of children with a cochlear implant). Logopedie en Foniatrie 2011; 83: 112–118.
- Zimmerman-Phillips S, Robbins AM and Osberger MJ. Assessing cochlear implant benefit in very young children. *Ann Otol Rhinol Laryngol Suppl* 2000; 185: 42–43.
- Archbold S, Lutman ME and Marshall DH. Categories of auditory performance. *Ann Otol Rhinol Laryngol Suppl* 1995; 166: 312–314.

Visit Sage journals online journals.sagepub.com/ home/trd

Sage journals